

© The Author, 2024. Published by Cambridge University Press on behalf of British Institute of International and Comparative Law. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted re-use, distribution and reproduction, provided the original article is properly cited.

CONTROL MECHANISMS IN MULTILATERAL INVESTMENT TRIBUNALS: NAVIGATING PROCEDURAL MULTILATERALISM AND SUBSTANTIVE BILATERALISM

JOSHUA PAINE (D)

University of Bristol Law School, University of Bristol, Bristol, UK Email: joshua.paine@bristol.ac.uk

Abstract A future multilateral investment court (MIC) or multilateral appellate mechanism (MAM) will operate on a plurilateral basis, among States that become parties to the tribunal's constitutive instrument and grant it jurisdiction over disputes under their investment treaties. The creation of a MIC or MAM would involve a significant strengthening and centralization of dispute settlement institutions in the investment treaty regime, which is already overly dependent on law-development by adjudicators, reflected in well-established concerns about loss of State control. Thus, a key challenge in designing a MIC or MAM is to incorporate appropriate control mechanisms that will enable State input, without unduly undermining a MIC or MAM's independence. This article analyses control mechanisms in a MIC or MAM, considering a wide range of questions of institutional design. It highlights two fundamental tensions. One is the tension between independence and accountability. The other tension is between procedural multilateralism and substantive bilateralism. While the procedural law in a MIC or MAM will have been multilateralized, the substantive law the tribunal will interpret and apply will remain contained in mostly bilateral investment treaties, controlled by the parties to those agreements. This article addresses the challenges of designing a multilateral tribunal for a regime that lacks multilateral substantive law and contributes to wider debates over striking an appropriate balance between international judicial independence and Member State control.

Keywords: investor–State dispute settlement reform, United Nations Commission on International Trade Law (UNCITRAL) Working Group III, Multilateral Investment Court, Multilateral Appellate Mechanism, independence and accountability of international tribunals, non-disputing Party submissions, intervention, authoritative interpretations.

[ICLQ vol 73, October 2024 pp 819–851]

doi:10.1017/S0020589324000368

I. INTRODUCTION

A multilateral investment court (MIC) or multilateral appellate mechanism (MAM) for investment disputes is no longer an abstract or academic proposition. Draft provisions on the design of a standing, two-tier multilateral mechanism for the resolution of investment disputes (referred to in this article as a MIC)—which may include as one component a standing appellate tribunal that States can opt into without accepting the first-instance tribunal (referred to as a MAM)—have been discussed at the United Nations Commission on International Trade Law (UNCITRAL) Working Group III in recent years and remain on the Working Group's agenda. Both a MIC and a MAM are potential outcomes of the Working Group III process and both are likely to operate on a plurilateral opt-in basis as they are only supported by certain States. Specifically, it has been clear since the early days of Working Group III that there is a subset of States—most notably the European Union (EU) and its Member States—that favour the creation of a two-tier MIC,² and another, potentially broader, group of States, including China and Morocco, which favour a standing MAM to sit above the existing system of ad hoc arbitration.³ As noted above, the appellate function could be performed by the appellate tier of a MIC if a so-called 'open architecture' is pursued

¹ See especially UNCITRAL, 'Draft Statute of a Standing Mechanism for the Resolution of International Investment Disputes: Note by the Secretariat' (8 February 2024) UN Doc A/CN.9/ WG.III/WP.239 (Draft Statute of a Standing Mechanism); UNCITRAL, 'Annotations to the Draft Statute of a Standing Mechanism for the Resolution of International Investment Disputes: Note by the Secretariat' (10 February 2024) UN Doc A/CN.9/WG.III/WP.240 (Annotations to Draft Statute). For discussion of these papers, see UNCITRAL, 'Report of Working Group III (Investor-State Dispute Settlement Reform) on the Work of Its Forty-Ninth Session (Vienna, 23-27 September 2024)' (16 October 2024) UN Doc A/CN.9/1194, paras 13-56 (Report of Working Group III Forty-Ninth Session); UNCITRAL, 'Report of Working Group III (Investor-State Dispute Settlement Reform) on the Work of its Forty-Eighth Session (New York, 1-5 April 2024)' (19 April 2024) UN Doc A/CN.9/1167, paras 84-112 (Report of Working Group III Forty-Eighth Session). For earlier working papers on these issues, see UNCITRAL, 'Standing Multilateral Mechanism: Selection and Appointment of ISDS Tribunal Members and Related Matters: Note by the Secretariat' (8 December 2021) UN Doc A/CN.9/WG.III/WP.213 (Standing Multilateral Mechanism: Note by the Secretariat); UNCITRAL, 'Appellate Mechanism: Note by the Secretariat' (17 November 2022) UN Doc A/CN.9/WG.III/WP.224.

² UNCITRAL, 'Possible Reform of Investor–State Dispute Settlement (ISDS): Submission from the European Union and its Member States' (24 January 2019) UN Doc A/CN.9/WG.III/WP.159/Add.1, paras 13–14. Several of the EU's treaty partners have agreed to pursue the establishment of a MIC and to refer investor–State disputes under the relevant IIA to a MIC if the latter is established: see, eg, Comprehensive Economic and Trade Agreement between Canada and the European Union (CETA) art 8.29; EU–Singapore Investment Protection Agreement (EU–Singapore IPA) art 3.12. Obviously, the EU is not a State and it is likely that the Statute of a MIC or MAM will be drafted so that is open to certain entities other than States, such as Regional Economic Integration Organizations or territories with independent powers: see Draft Statute of a Standing Mechanism ibid, art 41(1). M Bungenberg and A Reinisch, *Draft Statute of the Multilateral Investment Court* (Nomos 2021) 17, 49 (art 4(2)). Nevertheless, the article will refer to the State Parties to a MIC or MAM as a convenient shorthand, while recognizing that the Parties to a MIC or MAM may extend beyond States.

³ UNCITRAL, 'Possible Reform of Investor-State Dispute Settlement (ISDS): Submission from the Government of China' (19 July 2019) UN Doc A/CN.9/WG.III/WP.177, 4;

whereby States can opt in to a MIC's appellate jurisdiction only. While a MIC and a MAM are distinct reform proposals, this article treats both options together because they raise common questions in relation to the themes that are the focus of this article: ensuring State control and designing a multilateral adjudicatory institution for a regime where the substantive law will remain largely bilateral.

The creation of a MIC or MAM (or both) would constitute a significant strengthening and centralization of dispute settlement mechanisms in the investment treaty regime.⁵ Yet, as Alschner has noted, the investment treaty regime is already overly dependent on law-development by arbitral tribunals, reflected in well-established concerns about loss of State control.⁶ Accordingly, a further strengthening and centralization of adjudication, without an associated strengthening of mechanisms for State input and control, is unlikely to prove sustainable.⁷ Thus, as Yu notes in passing, if States choose to create a MIC or MAM (or both), steps should be taken in the design of the tribunal to address the significant risks of backlash arising from the new centralized tribunal's increased law-making influence.⁸

This article considers the design of appropriate control mechanisms in a MIC or a MAM. In a nutshell, a MIC or MAM needs to include control mechanisms that provide adequate opportunities for State input and control over the tribunal and its influence on the investment treaty regime. It is only with adequate avenues for State input and control that there may be long-term political buyin to the jurisdiction of a MIC or MAM. However, a MIC or MAM will also

UNCITRAL, 'Possible Reform of Investor–State Dispute Settlement (ISDS): Submission from the Government of Morocco' (11 February 2020) UN Doc A/CN.9/WG.III/WP.195, 3–6.

⁴ See, eg, Draft Statute of a Standing Mechanism (n 1) art 18(1); UNCITRAL, 'Summary of the Inter-Sessional Meeting on Investor–State Dispute Settlement (ISDS) Reform Submitted by the Government of Singapore' (20 October 2023) UN Doc A/CN.9/WG.III/WP.233, para 116 (Summary of the Inter-Sessional Meeting Submitted by the Government of Singapore). Alternatively, a MAM may be pursued independently of any MIC: UNCITRAL, 'Report of Working Group III (Investor–State Dispute Settlement Reform) on the Work of its Forty-Fourth Session (Vienna, 23–27 January 2023) '(7 February 2023) UN Doc A/CN.9/1130, para 122.

⁵ On the notion of the 'investment treaty regime', see J Bonnitcha, LNS Poulsen and M Waibel, *The Political Economy of the Investment Treaty Regime* (OUP 2017) 2–7.

⁶ W Alschner, Investment Arbitration and State-Driven Reform: New Treaties, Old Outcomes (OUP 2022) 10.

⁷ ibid 10, 278. Similarly, Yu argues that given the lack of shared understanding regarding substantive standards of investment protection among relevant actors, the creation of a MIC or MAM risks 'over-institutionalization', whereby the tribunal's jurisprudence may diverge from actors' understandings, leading to backlash: see C Yu, *Dispute Settlement and the Reform of International Investment Law: Legalization through Adjudication* (Edward Elgar 2023) 1–2, 94, 131, 163, 177–80. See also M Feldman, 'Investment Arbitration Appellate Mechanism Options: Consistency, Accuracy, and Balance of Power' (2017) 32 ICSID Rev 528, 530, 534–5, 538–44; J Wong and J Yackee, 'Transparency, Accountability, and Influence in the International Investment Law System', forthcoming, MichJIntlL, version of 4 November 2024 (on file with author) 53–4.
⁸ Yu ibid 164, 180.

⁹ Similarly, JE Alvarez, 'ISDS Reform: The Long View' (2022) 36 ICSID Rev 253, 272. On the function of control mechanisms in achieving and maintaining State buy-in, see JK Cogan, 'Competition and Control in International Adjudication' (2008) 48 VaJIntlL 411, 415, 419.

require substantial independence from the States Parties that create it if it is to fulfil the functions expected of it, such as interpreting international investment agreements (IIAs) in a principled manner which takes account of both State and investor interests. 10 Thus, as Roberts and St John argue, one 'polarity' or tension that must be 'managed' on an ongoing basis by policymakers in investor-state dispute settlement (ISDS) reform processes—and particularly in designing a MIC or MAM—is a 'tension between independence and accountability'. 11 It will be shown throughout this article that the tension between independence and accountability underlies numerous questions concerning the design of a MIC or MAM. While a MIC or MAM should be designed to include adequate mechanisms for State control, to ensure responsiveness to Member States and long-term political buy-in to the institution, it is also important that such control mechanisms are designed with safeguards against their abuse in mind. Although there is an existing literature on control mechanisms in the investment treaty regime, as Roberts and St John note, this largely focuses on the exercise of control mechanisms on a unilateral or bilateral basis, and how States can effectively exercise control in a multilateral setting is less well understood. 12

In considering the design of control mechanisms for a MIC or a MAM, this article gives particular attention to issues arising from the fact that while the procedural law in a MIC or MAM will have been multilateralized in the Statute of the tribunal and its procedural rules, the substantive law that the tribunal will apply will be contained in mostly bilateral investment treaties (BITs). At one level this simply reflects the procedural focus of the UNCITRAL Working Group III process, whose mandate intentionally focuses on reform of ISDS procedures rather than on the substantive standards in IIAs.¹³ Furthermore, as Titi notes, the phenomenon of

¹⁰ This article uses the terms 'international investment agreements' (IIAs) and 'investment treaties' as a shorthand that refers to bilateral investment treaties (BITs) and other treaties that often provide for a similar set of investment protections and may include consent to investor–State adjudication, particularly investment chapters of preferential trade agreements, eg, Bonnitcha, Poulsen and Waibel (n 5) 3–4.

¹¹ A Roberts and T St John, 'Complex Designers and Emergent Design: Reforming the Investment Treaty System' (2022) 116 AJIL 96, 138–9.

¹² ibid 108; see, eg, A Kulick (ed), Reassertion of Control over the Investment Treaty Regime (CUP 2016); A van Aaken, 'Control Mechanisms in International Investment Law' in Z Douglas, J Pauwelyn and JE Viñuales (eds), The Foundations of International Investment Law: Bringing Theory into Practice (OUP 2014); R Polanco, The Return of the Home State to Investor–State Disputes: Bringing Back Diplomatic Protection? (CUP 2019).

¹³ See, eg, United Nations General Assembly, 'Report of the United Nations Commission on International Trade Law Fiftieth Session' (3–21 July 2017) UN Doc A/72/17, paras 244–264. While there is debate over whether Working Group's III's mandate is limited to procedural issues, and how to draw the procedure/substance divide, most actors do not view the Working Group's mandate as covering substantive standards of investment protection: see, eg, J Bonnitcha et al, 'Damages and ISDS Reform: Between Procedure and Substance' (2023) 14(2) JIDS 213, 214–15; G Ünüvar, 'The Mandate Conundrum: Reflections on the 46th Session of the UNCITRAL Working Group III on ISDS Reform' (*EJIL Talk!*, 21 November 2023)

procedural multilateralism—despite the substantive law remaining bilateral—is not entirely new in investment law, as illustrated by procedurally focused multilateral instruments such as the Convention on the Settlement of Investment Disputes between States and Nationals of Other States (ICSID Convention) and the United Nations (UN) Convention on Transparency in Treaty-based Investor-State Arbitration. 14 Yet such an approach means that certain control mechanisms will remain solely in the hands of the parties to the relevant IIA—for example, there is little doubt that adopting authoritative interpretations of an IIA, treaty amendments, or terminating an IIA will remain control mechanisms exercisable exclusively by the parties to the relevant IIA.¹⁵ In contrast, other control mechanisms are likely to involve all States Parties to the Statute of a MIC or MAM—for example, electing the members of the tribunal, or adopting the tribunal's budget. Third-party involvement in proceedings before a MIC or MAM is an area where this tension between procedural multilateralism and substantive bilateralism will be particularly acute. For example, should States that are parties to the Statute of a MIC or MAM, but not parties to the particular IIA invoked in a dispute, be given some ability to make submissions in disputes that raise issues concerning interpretation of the tribunal's Statute, or that raise issues of wider systemic importance for numerous other IIAs? Given one of the aims of creating a MIC or MAM is for the standing tribunal to provide greater consistency in the interpretative approach applied to IIAs, 16 it is foreseeable that initial positions taken by a MIC or MAM, for example regarding terms or concepts contained in numerous IIAs, will prove influential in later cases concerning other IIAs.17

< https://www.ejiltalk.org/the-mandate-conundrum-reflections-on-the-46th-session-of-the-uncitral-working-group-iii-on-isds-reform/>.

¹⁴ C Titi, 'Procedural Multilateralism and Multilateral Investment Court: Discussion in Light of Increased Institutionalism in Transatlantic Relations' in E Fahey (ed), Institutionalisation beyond the Nation State Transatlantic Relations: Data, Privacy and Trade Law (Springer 2018) 150; Convention on the Settlement of Investment Disputes between States and Nationals of Other States (adopted 18 March 1965, entered into force 14 October 1966) 575 UNTS 159 (ICSID Convention); UN Convention on Transparency in Treaty-based Investor–State Arbitration (adopted 10 December 2014, entered into force 18 October 2017) 3208 UNTS.

¹⁵ See, eg, Submission from the European Union (n 2) paras 26–27.

16 ibid, paras 41–42; M Bungenberg and A Reinisch, From Bilateral Arbitral Tribunals and Investment Courts to a Multilateral Investment Court: Options Regarding the Institutionalization of Investor–State Dispute Settlement (2nd edn, Springer 2020) 18–19, 23.

¹⁷ See, eg, G Kaufmann-Kohler and M Potestà, 'Can the Mauritius Convention Serve as a Model for the Reform of Investor–State Arbitration in Connection with the Introduction of a Permanent Investment Tribunal or an Appeal Mechanism? Analysis and Roadmap' (Center for International Dispute Settlement, 3 June 2016) paras 73, 188 https://ssrn.com/abstract=345551]>. Contrast: D McRae, 'The WTO Appellate Body: A Model for an ICSID Appeals Facility?' (2010) 1 JIDS 371, 382–6 (acknowledging there are common standards and concepts in investment treaties but arguing that an appellate mechanism pursuing systemic consistency would be illegitimate, as each IIA must be interpreted according to its own terms); and B Legum, 'Appellate Mechanisms for Investment Arbitration: Worth a Second Look for the Trans-Pacific Partnership and the

This article's contribution lies not only in its analysis of a wide range of questions of institutional design of a MIC or MAM, but also in its elaboration of a conceptual framework for understanding two fundamental tensions that underlie and animate many of the specific questions of institutional design. These two recurrent tensions are those introduced above: the tension between independence and accountability or State control; and the tension between procedural multilateralism and substantive bilateralism. Through its analysis of control mechanisms in a future MIC or MAM, this article contributes to broader debates in international (economic) law of how to balance the independence of international tribunals with control by, and responsiveness to the preferences of, Member States. 18

This article proceeds as follows. Section II analyses several key concepts on which the article relies, namely mechanisms for State control in international tribunals and the concepts of independence and accountability of international tribunals. Section III considers those control mechanisms that are likely to operate on a multilateral basis, involving all States Parties to a MIC or MAM. Section IV addresses control mechanisms that are likely to remain solely in the hands of the parties to the particular IIA at issue. Section V then turns to control mechanisms that raise the tension between procedural multilateralism and the substantive law that a MIC or MAM will apply being largely contained in bilateral treaties controlled by the parties to those agreements. Among other issues, Section V considers approaches to third-State participation in proceedings before a MIC or MAM. Section VI concludes.

II. DESIGNING CONTROL MECHANISMS

Thinking about the design of a MIC or MAM does not start from a blank slate. Rather, it can rely on a significant existing literature and experience in other contexts concerning mechanisms for State control in international tribunals. ¹⁹ This section reviews existing knowledge and experience regarding control mechanisms in international tribunals, which can inform debates about the design of a MIC or MAM. It also reviews the concepts of judicial independence and judicial accountability, as these are key values which must be balanced in designing control mechanisms.

Proposed EU–US FTA?' in JE Kalicki and A Joubin-Bret (eds), Reshaping the Investor-State Dispute Settlement System: Journeys for the 21st Century (Brill Nijhoff 2015) 438 (similar).

¹⁸ See also Y Fukunaga, 'Precedent in Investment Arbitration: Is an Institutionalized Investment Court More Desirable?' in CJ Cheng (ed), *A New Global Economic Order: New Challenges to International Trade Law* (Brill Nijhoff 2022) 339; M Langford, CD Creamer and D Behn, 'Regime Responsiveness in International Economic Disputes' in S Gáspár-Szilágyi, D Behn and M Langford (eds), *Adjudicating Trade and Investment Disputes: Convergence or Divergence?* (CUP 2020). See also text accompanying nn 60–61.

¹⁹ SW Schill and G Vidigal, 'Designing Investment Dispute Settlement à la Carte: Insights from Comparative Institutional Design Analysis' (2019) 18 LPICT 314, 315–16.

A. Mechanisms for State Control in International Tribunals

'Control' has been defined in this context as 'checks on the powers of an [international] organization that ensure that the organization acts within its assigned mandate'.²⁰ This subsection does not purport to introduce every possible control mechanism, but to introduce those most relevant to a MIC or MAM, which will then be analysed in subsequent sections.²¹ At the outset, it is worth noting one common consideration that applies to most of the control mechanisms discussed below, which is directly relevant to ongoing policy debates over designing a MIC or MAM. In short, because many control mechanisms require States to act as a collective principal—ie to coordinate and reach agreement, which is often difficult, particularly in a multilateral context due to States' diverging preferences—the various means of State control over international tribunals are less effective than might be anticipated.²²

One well-recognized control mechanism is precision in drafting the substantive law that a tribunal applies. ²³ As an *ex ante* control mechanism this tool is familiar in the investment treaty regime, in the form of States' attempts to draft more precise IIAs over the last 20 years. ²⁴ In the investment treaty regime, adding greater precision *ex post* takes the form of authoritative interpretations agreed by treaty parties, agreements to amend the relevant IIA, or agreements to negotiate an entirely new IIA to replace the existing treaty. A related control mechanism known in other international tribunals is specifying a particular interpretative approach that the tribunal must follow. ²⁵ A well-known example is Article 3.2 of the World Trade Organization's (WTO) Dispute Settlement Understanding (DSU), which provides that one function of WTO dispute settlement is to clarify the provisions of WTO-covered agreements 'in accordance with customary rules of interpretation of public international

²⁰ Cogan (n 9) 413.

²¹ For typologies of control mechanisms in international tribunals, see ibid 418–20; LR Helfer and AM Slaughter, 'Why States Create International Tribunals: A Response to Professors Posner and Yoo' (2005) 93 CalLRev 899, 942–54; LR Helfer, 'Why States Create International Tribunals: A Theory of Constrained Independence' in S Voigt, M Albert and D Schmidtchen (eds), *International Conflict Resolution* (Mohr Siebeck 2006) 263–74. On the concept of control mechanisms in international tribunals, see also E Voeten, 'International Judicial Independence' in JL Dunoff and MA Pollack (eds), *Interdisciplinary Perspectives on International Law and International Relations: The State of the Art* (CUP 2012) 422–36; MA Pollack, 'International Relations Theories of Adjudication' in H Ruiz-Fabri (ed), *Max Planck Encyclopedia of International Procedural Law* (OUP 2020) paras 47–48.

²² Cogan (n 9) 426–8. See also KJ Alter, 'Delegation to International Courts and the Limits of Re-Contracting Political Power' in DG Hawkins et al (eds), *Delegation and Agency in International Organizations* (CUP 2006) 314; MA Pollack, *The Engines of European Integration: Delegation, Agency, and Agenda Setting in the European Union* (OUP 2003) 43–4 (on multiple principals limiting the effectiveness of control mechanisms).

Helfer and Slaughter (n 21) 945; Cogan ibid 421; contrast Alter ibid 322–4.

²⁴ See generally Alschner (n 6); JK Sharpe, 'From Delegation to Prescription: Interpretive Authority in International Investment Agreements' in CN Brower et al (eds), *By Peaceful Means: International Adjudication and Arbitration – Essays in Honour of David D. Caron* (OUP 2024).

law' and that '[r]ecommendations and rulings of the DSB [Dispute Settlement Body] cannot add to or diminish the rights and obligations' of members. ²⁶ Other precedents also exist, for example the deferential standards of review adjudicators are instructed to apply under Article 17.6 of the WTO Anti-Dumping Agreement, ²⁷ or the instruction in the WTO Multi-Party Interim Appeal Arrangement (MPIA) that '[t]he arbitrators shall only address those issues that are necessary for the resolution of the dispute. They shall address only those issues that have been raised by the parties. ²⁸ While specifying a particular interpretative approach can have an important signalling effect for adjudicators, it is difficult for treaty negotiators through such *ex ante* formal legal regulation to control precisely how a tribunal will perform its functions, including as tribunals are typically left to interpret their own mandate. ²⁹

Another important category of control mechanisms concerns the appointment of members of an international tribunal. At the stage of drafting a constitutive instrument, choices facing treaty negotiators include the term length for appointments, whether to provide for renewable terms, which bodies elect judges and by what voting rules, the criteria specified for judges in terms of required expertise, geographic diversity and other forms of diversity, and whether a process for screening nominees for suitability prior to appointment is created.³⁰ At the stage of drafting a constitutive instrument, the States Parties to a tribunal exercise these control mechanisms collectively, as it is only with the agreement (or at least non-objection) of all treaty parties that such features will end up in a constitutive instrument. In contrast, at the stage of making (re)appointments to a tribunal, there may be some scope for control to be exercised by individual States parties, or a subset of States parties to the tribunal, for example if there is a requirement that candidates are nominated by a Member State of the tribunal or by a regional grouping, or if powerful States are able to exercise a veto.³¹ The last point highlights that the voting

²⁶ Understanding on Rules and Procedures Governing the Settlement of Disputes, Marrakesh Agreement Establishing the World Trade Organisation, Annex 2 (adopted 15 April 1994, entered into force 1 January 1995) 1869 UNTS 401 (WTO DSU) art 3.2.

²⁷ Agreement on Implementation of Article VI of the General Agreement on Tariffs and Trade 1994, Marrakesh Agreement Establishing the World Trade Organisation, Annex 1A (adopted 15 April 1994, entered into force 1 January 1995) 1868 UNTS 201 (WTO Anti-Dumping Agreement) art 17.6. On this provision, see, eg, JH Fahner, *Judicial Deference in International Adjudication: A Comparative Analysis* (Hart Publishing 2020) 66–9.

Multi-Party Interim Appeal Arbitration Arrangement pursuant to Article 25 of the DSU, JOB/DSB/1/Add.12, 30 April 2020, Annex 1, para 10.
 Cogan (n 9) 438–9.

³⁰ ibid 423; see generally O Larsson et al, 'Selection and Appointment in International Adjudication: Insights from Political Science' (2023) 14 JIDS 134; R Mackenzie et al, *Selecting International Judges: Principle, Process, and Politics* (OUP 2010).

³¹ See generally JL Dunoff and MA Pollack, 'The Judicial Trilemma' (2017) 111 AJIL 225, 235 (focusing on rules on judicial (re)appointment and differentiating between accountability of individual judges to an individual nominating State and accountability to the wider group of States Parties to a tribunal involved in electing judges). See also a recent study of the ICJ's electoral system which highlights that within regional groupings there has been a lack of rotation, with nationals of certain States appointed to the Court for long periods of time: MJA Oyarzabal,

rules applicable to a particular decision are a crucial element of institutional design and will affect whether a particular control mechanism operates collectively, involving all States Parties to a tribunal, or gives rise to some possibility for unilateral control by individual Member States or control by a subset of Member States. For example, if a particular decision (eg appointment of tribunal members) is made subject to a requirement of consensus, defined as no Member State objecting, this will mean that a control mechanism that operates multilaterally in practice gives rise to a significant risk of unilateral vetoes by individual Member States, and particularly by powerful Member States which are able to wield a threat of veto credibly.³² Interestingly, the UNCITRAL secretariat Draft Statute of a Standing Mechanism provides that where the Conference of the Contracting Parties—the plenary body of Member States—cannot make a decision by consensus, decisions may be made by a four-fifths majority of Contracting Parties present and voting.³³ This approach, if adopted, would eliminate the risk of unilateral vetoes and substantially reduce the risk of the tribunal being held hostage by a small minority of Contracting Parties.

A third important mechanism for State control is a tribunal's budget.³⁴ As Cogan notes, international 'courts are entirely dependent on States and international organizations for their funding'.³⁵ Most international tribunals are primarily funded from the assessed budgetary contributions of the Member States—either the Member States of the relevant international tribunal (as in the case of the International Criminal Court (ICC)) or the Member States of the parent international organization where the tribunal is funded from the budget of a parent organization (as in the case of the International Court of Justice (ICJ) and the WTO Appellate Body (AB)).³⁶ Typically the budget must be adopted by the plenary political body that oversees the relevant international tribunal, eg the UN General Assembly for the ICJ, or the Assembly of States Parties (ASP) for the ICC.³⁷ While a tribunal's budget may formally need to be approved by the plenary body of States that governs a tribunal, the issue of budgetary control also highlights that certain control mechanisms are in practice more available to some

^{&#}x27;Election of Judges to the International Court of Justice: Proposals for Reform without Amending the Statute' (2024) 73 ICLQ 361, 365–7.

³² In the General Agreement on Tariffs and Trade (GATT)/WTO context other commentators have argued that the rule of consensus decision-making does not give each member equal power, but 'resembles weighted voting', where large, powerful members are better able to wield the threat of veto: see J Pauwelyn, 'The Transformation of World Trade' (2005) 104 MichLRev 1, 43–4; RH Steinberg, 'In the Shadow of Law or Power? Consensus-Based Bargaining and Outcomes in the GATT/WTO' (2002) 56 IntlOrg 339, esp 346–9. See also RH Steinberg, 'Judicial Lawmaking at the WTO: Discursive, Constitutional, and Political Constraints' (2004) 98 AJIL 247, 249, 264–5.

³³ Draft Statute of a Standing Mechanism (n 1) art 4(7)–(8).

³⁴ Helfer and Slaughter (n 21) 948; Cogan (n 9) 423–4. See generally T Ingadottir, 'The Financing of International Adjudication' in CPR Romano, KJ Alter and Y Shany (eds), *The Oxford Handbook of International Adjudication* (OUP 2013) esp 595, 600, 610.

35 Cogan ibid 423.

36 Ingadottir (n 34) 600–1.

37 ibid 608–9.

Member States than others.³⁸ For example, a threat to withhold paying assessed budgetary contributions is obviously far more serious for a tribunal when it comes from a major donor.³⁹

A fourth category of control mechanisms concerns the design of rules concerning the jurisdiction of a tribunal. 40 Most obviously this concerns which kinds of actors are given access to a tribunal as parties, whether an additional act of State consent is needed for the tribunal to have jurisdiction besides the constitutive instrument creating the tribunal, and which kinds of disputes a tribunal is given jurisdiction over (eg what subject matter limitations are imposed). Carve-outs of sensitive issues from any advance grant of State consent to adjudication are an important control mechanism that treaty drafters enjoy. 41 It is widely recognized that tribunals that confer standing on non-State actors are much harder for States to control than tribunals that only permit State—State claims, given the likelihood of a far higher volume of cases in the former scenario and the loss of State control over which cases are brought before a tribunal. 42

A fifth broad category of control mechanism concerns the procedural law to be applied by a tribunal. While many constitutive instruments delegate to an international tribunal the power to determine its own procedures, ⁴³ alternatives exist granting States greater control. An example is the Rome Statute of the ICC, which provides that the Rules of Procedure and Evidence (and amendments thereto) must be adopted by a two-thirds majority of members of the ASP. ⁴⁴ While there is provision for the ICC judges, by two-thirds majority, to adopt provisional rules where the Rules of Procedure 'do not provide for a specific situation before the Court', these can be adopted, amended or rejected by the ASP. ⁴⁵ Although the Rome Statute provides that the ICC judges shall adopt 'the Regulations of the Court necessary for its routine functioning' by majority and these take effect upon adoption, they must be circulated to States Parties for comments and only remain in force if there is not an objection within six months from a majority of States Parties. ⁴⁶

³⁸ See also text accompanying n 32.

³⁹ See Ingadottir (n 34) 609, noting 'in practice, some states have more budgetary power than others. At the United Nations, with respect to budgetary decisions, for the past 20 years there has been an informal understanding that the biggest contributors have a greater say in the discussion and adoption of the budget, and at times a *de facto* veto right.'

⁴⁰ Cogan (n 9) 420.

⁴¹ Helfer and Slaughter (n 21) 945.

⁴² See, eg, Helfer and Slaughter ibid 952–3; RO Keohane, A Moravcsik and A-M Slaughter, 'Legalized Dispute Resolution: Interstate and Transnational' (2000) 54 IntlOrg 457, 462–6.

⁴³ See, eg, Statute of the International Court of Justice (adopted 26 June 1945, entered into force 24 October 1945) XV UNCIO 355 (ICJ Statute) art 30(1); Statute of the International Tribunal for the Law of the Sea, UN Convention on the Law of the Sea (adopted 10 December 1982, entered into force 16 November 1994) annex VI (ITLOS Statute) art 16.

 ⁴⁴ Rome Statute of the International Criminal Court (adopted 17 July 1998, entered into force 1 July 2002) 2187 UNTS 90 (Rome Statute of the ICC) art 51(1)–(2).
 45 ibid, art 51(3).
 46 ibid, art 52.

A sixth potential control mechanism lies in the provisions governing the secretariat of an international tribunal. At one level, a secretariat can be a means of controlling adjudicators and the potential for rogue interpretations.⁴⁷ However, significant concerns have been raised about the oversized and unaccountable influence of secretariats, particularly in WTO dispute settlement.⁴⁸ Such concerns have clearly influenced thinking about the design of a secretariat for a MIC or MAM, for example the provision in the current Draft Statute for term limits for the Executive Director of the Secretariat and clarification that the Executive Director 'shall be accountable to the Conference [of the Contracting Parties]'.⁴⁹

A seventh group of control mechanisms concerns the kinds of remedies a tribunal is authorized to award, the legal status of decisions rendered, and the practical scope for non-compliance by States.⁵⁰ Constitutive instruments can limit the scope of remedies open to a tribunal. One example is the system of remedies under the WTO DSU, which removes the potential for an award of compensation to redress past wrongful conduct.⁵¹ Another is the common specifications in newer investment treaties that if a tribunal orders restitution of property, it must provide an alternative to the State of paying compensation in lieu of restitution, and that a tribunal may not award punitive damages.⁵² While most constitutive instruments provide that a tribunal's decisions are binding for the disputing parties as a matter of international law, there are important variations regarding whether tribunals' decisions are made enforceable in domestic legal systems.⁵³ For present purposes, the key point is that in settings where judgments of international tribunals are enforceable in domestic legal systems, this reduces State control over the system.⁵⁴ The existing system of investor–State arbitration sits towards the strong end of the remedies/enforcement spectrum, as tribunals' decisions are enforceable in domestic systems pursuant to powerful multilateral frameworks, principally the ICSID Convention and the UN Convention on the Recognition and Enforcement of Foreign Arbitral Awards. 55

⁴⁷ See, eg, J Kucik and S Puig, 'Towards an Effective Appellate Mechanism for ISDS Tribunals' (2023) 22 WorldTR 562, 578.

⁴⁸ See, eg, J Pauwelyn and K Pelc, 'Who Guards the "Guardians of the System"? The Role of the Secretariat in WTO Dispute Settlement' (2022) 116 AJIL 534.

⁴⁹ Draft Statute of a Standing Mechanism (n 1) art 6(2)–(3).

⁵⁰ Helfer and Slaughter (n 21) 946, 952.

⁵¹ See, eg, C Brown, A Common Law of International Adjudication (OUP 2007) 218–20.

⁵² See, eg, North American Free Trade Agreement (signed 17 December 1992, entered into force 1 January 1994, terminated 1 July 2020) (NAFTA) art 1135(1)(b), (3); Dominican Republic-Central America Free Trade Agreement (signed 5 August 2004, entered into force 1 March 2006) (DR-CAFTA) art 10.26(1)(b), (3); CETA (n 2) art 8.39(1)(b), (4).

See generally Keohane, Moravcsik and Slaughter (n 42) 466–9, 475–8; JM Smith, 'The Politics of Dispute Settlement Design: Explaining Legalism in Regional Trade Pacts' (2000) 54
 IntlOrg 137, 140–2.
 Keohane, Moravcsik and Slaughter ibid 476–8.

⁵⁵ ICSID Convention (n 14) art 54. UN Convention on the Recognition and Enforcement of Foreign Arbitral Awards (adopted 10 June 1958, entered into force 7 June 1959) (New York Convention) arts III–V.

B. Independence and Accountability of International Tribunals

In the context of international tribunals, judicial independence is typically understood as a 'set of institutional and other factors that, to a lesser or greater extent ... allows judges to develop legal opinions unconstrained by the preferences of other actors'. ⁵⁶ Existing discussions largely focus on the independence of international tribunals from the States that create and maintain such institutions. ⁵⁷ For example, Keohane, Moravcsik and Slaughter defined independence as 'the extent to which adjudicators ... are able to deliberate and reach legal judgments independently of national governments', ⁵⁸ arguing:

[t]he extent to which members of an international tribunal are independent reflects the extent to which they can free themselves from at least three categories of institutional constraint: selection and tenure, legal discretion, and control over material and human resources.⁵⁹

It is important to remember that judicial independence is not an absolute value, but is balanced against other competing interests,⁶⁰ which include responsiveness to the States that create and sustain an international tribunal.⁶¹

Judicial accountability is 'in many ways merely the other side of the coin to judicial independence'. Accountability is often understood as the principle whereby 'some actors have the right to hold other actors to a set of standards, to judge whether they have fulfilled their responsibilities in light of these standards, and to impose sanctions if they determine that these responsibilities have not been met'. Wong and Yackee emphasize that the point of the sanctioning involved in accountability is ultimately to influence the actor being held to account and to incentivize that actor to act in future consistently with the preferences of the actor doing the holding to account. While a variety of actors could have a role in holding international tribunals accountable, in terms of control mechanisms what is of interest is

Voeten (n 21) 421–2. See also P Mahoney, 'The International Judiciary – Independence and Accountability' (2008) 7 LPICT 313, 322–3; H Keller and S Meier, 'Independence and Impartiality in the Judicial Trilemma' (2017) 111 AJILUnbound 344, 345–6; to a limited extent, Subsection II(B) draws on J Paine, 'The WTO's Dispute Settlement Body as a Voice Mechanism' (2019) 20 JWorldInv&Trade 820, 834–5.

⁵⁸ Keohane, Moravcsik and Slaughter (n 42) 459–60.

⁵⁹ ibid 460. See generally 459–62, 470–2.

⁶⁰ A Seibert-Fohr, 'International Judicial Ethics' in Romano, Alter and Shany (n 34) 774.

⁶¹ Y Shany, Assessing the Effectiveness of International Courts (OUP 2014) 109–10; Voeten (n 21) 438–9 ('there may be politically optimal levels of judicial independence'); Helfer and Slaughter (n 21) 942–3.

⁶² Mahoney (n 56) 347–8; Larsson et al (n 30) 135 ('independence and accountability are in conflict with each other: the more independent judges are, the less accountable they will be, and vice versa').

⁶³ RW Grant and RO Keohane, 'Accountability and Abuses of Power in World Politics' (2005) 99 AmPolSciRev 29, 29–30. Cited in Dunoff and Pollack (n 31) 233.

65 See, eg, Mahoney (n 56) 339.

https://doi.org/10.1017/S0020589324000368 Published online by Cambridge University Press

accountability to the States that create and sustain an international tribunal.⁶⁶ As noted above, it is possible to distinguish between control mechanisms that are exercised collectively by all parties to a tribunal's constitutive instrument, and control mechanisms that may be open to individual States Parties, or a subset of States Parties to the constitutive instrument.⁶⁷

III. MULTILATERAL CONTROL MECHANISMS

This section considers those control mechanisms that will likely be exercised by all States Parties to the statute of a MIC or MAM. The UNCITRAL secretariat Draft Statute of a Standing Mechanism includes the creation of a plenary body, 'the Conference of the Contracting Parties', which is composed of representatives of all Parties to the instrument establishing the Standing Mechanism, and is given broad responsibility to oversee the functioning of the Standing Mechanism.⁶⁸ The creation of such a plenary body follows the example of other 'international judicial governance institutions', for example the DSB in relation to WTO panels and the AB, the ASP in relation to the ICC, or the Meeting of States Parties to the UN Convention on the Law of the Sea (SPLOS) in relation to the International Tribunal for the Law of the Sea (ITLOS).⁶⁹

A. Appointment of MIC or MAM Members

It is likely that all parties to a MIC or MAM will collectively exercise control over the appointment of members to the tribunal or tribunals. However, as explained above, much will depend upon the decision-making rules that apply. The UNCITRAL secretariat Draft Statute of a Standing Mechanism gives the Conference of the Contracting Parties—the plenary body of all Member States—the role of appointing the members of the first-instance tribunal and appeals tribunal, and of establishing a Selection Committee that has the role of screening nominees to determine whether they meet the qualification requirements, prior to the Conference making appointments to the tribunals. From the perspective of State control, it is notable that within the Working Group III process there appears to be substantial support for a screening mechanism, with the purpose of screening out candidates that do not meet the relevant criteria, in between the nomination of candidates and

67 See text accompanying nn 31–32.

⁶⁶ See, eg, N Blokker, 'The Governance of International Courts and Tribunals: Organizing and Guaranteeing Independence and Accountability' in A Follesdal and G Ulfstein (eds), *The Judicialization of International Law: A Mixed Blessing?* (OUP 2018) 38–40.

⁶⁸ Draft Statute of a Standing Mechanism (n 1) arts 3(1)–(2), 4; Standing Multilateral Mechanism: Note by the Secretariat (n 1) para 15 (draft provision 3(a)(1)).

Blokker (n 66) 27, 33–5.
 Draft Statute of a Standing Mechanism (n 1) arts 4(2)(b), 11.
 See text accompanying nn 32–33.
 Draft Statute of a Standing Mechanism (n 1) arts 4(2)(b), 11.

their appointment.⁷³ This approach is clearly inspired by the screening mechanisms developed in recent years in certain other international tribunals.⁷⁴ Such a screening mechanism offers reduced State control compared to a system where States have complete discretion to judge the suitability of candidates. However, State control is not the only relevant value, and it may be that the parties to a MIC or MAM are willing to relinquish some discretion if it provides a greater guarantee of only qualified candidates proceeding to the election stage. Similarly, it is noteworthy that, following earlier State feedback, the UNCITRAL secretariat Draft Statute permits the Conference of the Contracting Parties to make an open call for candidates—ie to expand the range of actors that may nominate candidates beyond Member States.⁷⁵ Again, this would represent a weakening of State control—compared to a system only permitting nominations by Member States—in the interest of furthering other values (eg a transparent and depoliticized process for judicial appointments).⁷⁶

Another interesting design choice concerns whether all parties to the instrument establishing a MIC or MAM will be entitled to vote on the election of members, or only those States that accept the jurisdiction of the relevant tribunal. This will be a particular issue if an 'open architecture' is pursued whereby States can opt into different parts of the standing tribunal (eg only the appellate tribunal).⁷⁷ The question will then be, do States that have only accepted the appellate tribunal's jurisdiction have a right to participate in decisions concerning the first-instance level of the MIC (eg election of first-instance tribunal members)? An earlier UNCITRAL secretariat draft appeared to suggest that accepting the jurisdiction of a MIC need not be a requirement for parties to the MIC's statute to vote on the election of judges, noting 'there are several courts in which tribunal members are selected by treaty parties or by a collective body of States, even if that membership is larger than the group of States that accept the court's

⁷³ ibid. See discussion in Report of Working Group III Forty-Ninth Session (n 1) paras 49–56. See also Standing Multilateral Mechanism: Note by the Secretariat (n 1) para 41.

⁷⁴ See Standing Multilateral Mechanism: Note by the Secretariat ibid, para 42; Report of Working Group III Forty-Ninth Session ibid, para 51; N Tsereteli and H Smekal, 'The Judicial Self-Government at the International Level—A New Research Agenda' (2018) 19 GermLJ 2137, 2147–50; Larsson et al (n 30) 146–7; Oyarzabal (n 31) 373–9 (reviewing existing screening mechanisms and proposing such a mechanism for the ICJ).

⁷⁵ Draft Statute of a Standing Mechanism (n 1) art 9(2). See also art 10(5). See also the discussion of this issue in Report of Working Group III Forty-Ninth Session ibid, paras 40–47. For earlier consideration of this issue, see Standing Multilateral Mechanism: Note by the Secretariat ibid, paras 33–39, and the discussion reported in UNCITRAL, 'Report of Working Group III (Investor–State Dispute Settlement Reform) on the Work of its Forty-Second Session (New York, 14–18 February 2022)' (23 March 2022) UN Doc A/CN.9/1092, paras 63–70 (Report of Working Group III Forty-Second Session).

⁷⁶ See, eg, Report of Working Group III Forty-Second Session ibid, para 68; Report of Working Group III Forty-Ninth Session ibid, para 42.

⁷⁷ See, eg, Draft Statute of a Standing Mechanism (n 1) art 18(1); Standing Multilateral Mechanism: Note by the Secretariat (n 1) para 7.

jurisdiction', and citing examples such as the ICJ and ITLOS.⁷⁸ In comments on this draft, the EU objected to the possibility of members of the MIC being elected by States 'other than those that accept the tribunal's jurisdiction'.⁷⁹ The most recent secretariat documents acknowledge that this issue remains to be resolved.⁸⁰ If the ability to vote on electing members of a MIC or MAM is made conditional on a State's acceptance of the jurisdiction of the relevant tribunal, one question that will arise is what degree of acceptance of jurisdiction is required? For example, would a State's acceptance of the jurisdiction of the appellate tribunal for a single dispute, or for disputes under just one of its IIAs, mean that the State is entitled to vote on appellate tribunal appointments?

Staying with the election of members to the MIC or MAM, another interesting and yet-to-be-resolved design choice is whether all State Parties to the MIC or MAM would be entitled to vote on all nominated candidates, or whether the various regional groupings—seen as important to ensure geographic diversity—would only vote on candidates from their region.⁸¹ While an initial secretariat draft appeared to suggest the latter approach,⁸² in discussions in Working Group III 'it was generally felt that the members of the Committee of the Parties should be entitled to vote on all identified candidates and not limited to those that fall within their regional group',⁸³ and several States in written comments have opposed the idea of States only voting on candidates from their regional groups.⁸⁴ If voting is not confined to regional groups, then the control mechanism of deciding on judicial appointments will operate more diffusely, involving all States Parties to a MIC or MAM, assuming that there are more nominees from each regional group than seats allocated to the group.⁸⁵

⁷⁸ Standing Multilateral Mechanism: Note by the Secretariat ibid, para 32.

⁷⁹ Written comments of the EU on the initial secretariat draft 'Standing Multilateral Mechanism: Selection and Appointment of ISDS Tribunal Members and Related Matters', 10 <a href="https://uncitral.un.org/sites/uncitral

⁸⁰ See Annotations to Draft Statute (n 1) para 32.

⁸¹ ibid; Standing Multilateral Mechanism: Note by the Secretariat (n 1) paras 44–46, especially draft provision 8(3).

^{82*} Standing Multilateral Mechanism: Note by the Secretariat ibid, paras 44–46, especially draft provision 8(3); Bungenberg and Reinisch also propose that States Parties would be limited to voting on candidates from their regional group: Bungenberg and Reinisch (n 2) 54–5 (art 12(4)–(6)) and 23.

⁸³ UNCITRAL, 'Report of Working Group III (Investor–State Dispute Settlement Reform) on the Work of its Forty-Third Session (Vienna, 5–16 September 2022)' (7 October 2022) UN Doc A/ CN.9/1124, para 19.

⁸⁴ See, eg, UNCITRAL, Written comments of Canada, November 2021, 4, and Written comments of Singapore, 9, regarding the initial secretariat draft on 'Standing Multilateral Mechanism: Selection and Appointment of ISDS Tribunal Members and Related Matters' https://uncitral.un.org/en/multilateralpermanentinvestmentcourt.

⁸⁵ In other international tribunals there are examples of regional groups negotiating the withdrawal of candidates, so that the number of nominees is exactly the number of seats to be filled: see Mackenzie et al (n 30) 105–10.

B. Setting the Budget for a MIC or MAM

Another control mechanism that will likely be exercised by all States Parties to a MIC or MAM concerns the tribunal's budget. Given that only a subset of States support the reform option of creating a MIC or a MAM, it is unlikely that either body could be funded from the budget of another international organization (eg the UN budget).86 Rather, a MIC or MAM is likely to be funded from contributions of the States Parties to these bodies, similar to the ICC (funded from assessed contributions of the States Parties to the Rome Statute)87 or ITLOS (funded by the States Parties to the UN Convention on the Law of the Sea and the International Seabed Authority). 88 This could make the MIC or MAM relatively vulnerable to control in the form of funding cuts.

In the UNCITRAL secretariat Draft Statute, the Conference of the Contracting Parties is given broad control over the Standing Mechanism's funding. For example, the Conference is to '[a]dopt the annual budget of the Standing Mechanism' and to '[d]etermine the amount of remuneration of the members' of the first-instance and appeals tribunals.⁸⁹ The obvious challenge here will be to design the funding arrangements so that fixing the annual budget, including fixing the compensation of tribunal members, does not become a mechanism of political control by States Parties. 90 Reflecting this concern, one of the overarching general principles to govern the operation of the Standing Mechanism is that it shall be 'independent and free from undue external influence, including from its donors'. 91 An interesting precedent that could insulate a MIC or MAM from Member State pressure via a reduction of funding is the model of the Caribbean Court of Justice, where the Court is funded through the income earned on an independently managed trust fund, which was established by the States that established the Court to provide for the Court's running costs (including salaries).⁹² As Malleson has argued, this funding model may have 'important benefits in securing the long-term independence of ... [international] courts' and 'warrant[s] serious consideration for wider adoption by the community of international courts'. 93

A MIC or MAM may also be partly funded through the charging of user fees, similar to the administrative fees that disputing parties are charged by arbitral

⁸⁶ Compare Informal Draft prepared by the UNCITRAL Secretariat for the 6th intersessional meeting of Working Group III, 'Financing of a Standing Mechanism - An Outline', para 38 (Informal Draft on Financing of a Standing Mechanism) https://uncitral.un.org/sites/uncitral.un. org/files/media-documents/uncitral/en/financing_of_a_standing_mechanism_sept.2023.pdf>.

87 Rome Statute of the ICC (n 44) art 115.

Draft Statute of a Standing Mechanism (n 1) art 4(2)(j), (k).

⁹⁰ See, eg, G Kaufmann-Kohler and M Potestà, 'The Composition of a Multilateral Investment Court and of an Appeal Mechanism for Investment Awards' (CIDS, 15 November 2017) para 88 https://ssm.com/abstract=3457310.

91 Draft Statute of a Standing Mechanism (n 1) art 2(2).

⁹² See K Malleson, 'Promoting Judicial Independence in the International Courts: Lessons from 93 ibid 678. the Caribbean' (2009) 58 ICLQ 671, 677-8.

institutions. 94 Indeed, the UNCITRAL secretariat Draft Statute foresees the Conference of the Contracting Parties adopting a fee structure for the Standing Mechanism. 95 Obviously, charging user fees could reduce the MIC or MAM's dependence on Member State contributions. However, there is a clear concern in the Working Group III process—given the issues in the existing ISDS system that have motivated the reform process—to avoid any direct link between user fees, the volume or length of cases, and remuneration of members of a MIC or MAM.⁹⁶ Thus the UNCITRAL secretariat has noted that:

[i]t may be prudent that the budget for the operation of the Tribunals (in particular, the remuneration of the members of the Tribunals) relies only on contributions by Contracting Parties rather than on fees to be charged ... [to] ensure the independence and integrity of the Tribunals.97

C. Control over Procedural Law to be Applied by a MIC or MAM

Another control mechanism likely to be exercised by all States Parties to a MIC or MAM concerns the procedural law to be applied by the tribunal. As noted above, existing international tribunals demonstrate there are a variety of models, from delegating the creation of rules of procedure to the tribunal itself, to tribunals such as the ICC where Member States retain far greater control.98 The UNCITRAL secretariat Draft Statute gives the Committee of the Contracting Parties the power to adopt the rules of procedure of the firstinstance and appellate tribunals, and to '[a]dopt administrative, financial and other regulations on the operation of the Standing Mechanism'. 99 These provisions suggest that States Parties to a MIC or MAM are keen to retain control over the procedural law to be applied by the new tribunal. However, some form of residual power over procedures will need to be afforded to a MIC or MAM, given that the rules of procedure and other regulations adopted by the Committee of the Contracting Parties are unlikely to provide for every scenario that may arise. For example, in discussion of these aspects of an earlier UNCITRAL secretariat draft, 'it was suggested that flexibility should be given to the Tribunal to update its rules and adapt its procedure when necessary', 100 and several States highlighted the need for further thought regarding how to divide the power over procedures between the

⁹⁴ See, eg, Informal Draft on Financing of a Standing Mechanism (n 86) para 48; Bungenberg ⁹⁵ Draft Statute of a Standing Mechanism (n 1) art 4(2)(m). and Reinisch (n 16) 93.

⁹⁶ See, eg, Submission from the European Union (n 2) para 33; Informal Draft on Financing of a nding Mechanism (n 86) paras 48–49.

⁹⁷ Annotations to Draft Statute (n 1) para 101. Standing Mechanism (n 86) paras 48–49.

⁹⁸ See text accompanying nn 43–44.

Draft Statute of a Standing Mechanism (n 1) art 4(2)(f)(h). See also Standing Multilateral Mechanism: Note by the Secretariat (n 1) para 15 (draft provision 3(a)(3)).

Report of Working Group III Forty-Second Session (n 75) para 29. See also Report of Working Group III Forty-Eighth Session (n 1) para 93.

Tribunal and Member States.¹⁰¹ As noted above, even in the ICC, where the Rules of Procedure and Evidence are firmly in the control of the States Parties, the ICC judges are permitted to adopt provisional rules where the Rules of Procedure 'do not provide for a specific situation before the Court' and are also permitted to adopt 'the Regulations of the Court necessary for its routine functioning'.¹⁰²

D. Specifying an Interpretative Approach and the Functions of a MIC or MAM

Another set of control mechanisms that will be exercised collectively by all States involved in negotiating the Statute of a MIC or MAM concerns whether the Statute may specify an interpretative approach to be applied by the tribunal, and how it will specify and limit the functions of the tribunal. 103 For example, as one of the concerns about the creation of a MIC or MAM is that such a body may engage in developing a common approach across IIAs beyond what can be justified by the customary rules of treaty interpretation, ¹⁰⁴ States Parties may want to remind expressly a MIC or MAM in its Statute that it should take account of relevant differences between IIAs. 105 More aggressively, the designers of a MIC or MAM could seek to limit the precedential weight to be given to the tribunal's decisions, eg by providing that decisions shall only be treated as persuasive in later cases in relation to IIAs containing identical treaty language. 106 Likewise, if States Parties are concerned to limit the increased law-making power of a MIC or MAM they may also wish to borrow from relevant aspects of the MPIA, eg specifying that the tribunal shall not address issues that are not necessary for the resolution of the dispute. 107

Given persistent concerns over regulatory space in investment treaty arbitration, a MIC or MAM's statute might also codify a notion of deference, perhaps drawn from existing arbitral jurisprudence, eg instructing the MIC or MAM in assessing alleged breaches of an IIA to bear in mind the 'high level of deference that international law accords to Contracting Parties with regard to [the development of domestic policies as well as implementation of

¹⁰¹ See, eg, UNCITRAL (n 84) Written comments of Korea, 2, Written comments of Canada, 2, Written comments of Colombia, paras 7–8, Written comments of Switzerland, paras 2–3, regarding the initial secretariat draft on 'Standing Multilateral Mechanism: Selection and Appointment of ISDS Tribunal Members and Related Matters' (n 84).

See text accompanying nn 45–46.

On the latter point, see Kucik and Puig (n 47) 575, 579.

¹⁰⁴ See, eg, ibid 573; B Legum, 'Options to Establish an Appellate Mechanism for Investment Disputes' in KP Sauvant (ed), *Appeals Mechanism in International Investment Disputes* (OUP 2008) 235; NJ Calamita and C Giannakopoulos, *ASEAN and the Reform of Investor–State Dispute Settlement: Global Challenges and Regional Options* (Edward Elgar 2022) 140, 167; Alvarez (n 9) 272.

Summary of the Inter-Sessional Meeting Submitted by the Government of Singapore (n 4) para 62.
 See text accompanying n 28.

international commitments ...] [and] the right to regulate'.¹⁰⁸ In designing the appellate tier of a MIC, or a stand-alone MAM, States also face, and have begun working through, a series of choices that will limit (or expand) the potential for appeals, eg which types of decisions appeals are permitted in relation to, and whether appeals are limited to errors of law or also cover certain (eg manifest) errors in fact-finding.¹⁰⁹

E. Control over Enforcement of Decisions and Remedies before a MIC or MAM

It is likely that the statute of a MIC or MAM will be drafted so that, similar to the ICSID Convention, States Parties to the statute will be obliged to recognize awards of the tribunal as binding, and to enforce pecuniary remedies in their territories as if they were a final judgment of a domestic court. 110 This reflects the widely accepted view that for a MIC or MAM to be effective, and sufficiently attractive to investors, its decisions will need to be enforceable in domestic legal systems, similar to the existing system of investor–State arbitration. 111 Regarding the enforceability of decisions in non-Parties to a MIC or MAM, the decisions of a MIC or MAM may be enforceable under the New York Convention, 112 and it is also possible that non-Parties may agree to enforce decisions of the tribunal in their territories. 113 In short, a MIC or MAM is likely to remain at the strong end of the enforcement spectrum, thus underscoring the need for adequate mechanisms for State input and control.

If States were particularly concerned to limit the interpretative power delegated to a MIC or MAM, they could conceivably provide in the constitutive instrument that the decisions of the tribunal only become binding after they have been adopted by the Conference of the Contracting Parties. If this approach were pursued, much would depend on the voting rules applying to adoption of decisions. For example, if adoption of tribunal decisions were, as in GATT dispute settlement, subject to consensus this would give rise to unilateral vetoes—where a single State could block the adoption of

¹⁰⁸ UNCITRAL, 'Possible Reform of Investor–State Dispute Settlement (ISDS): Draft Provisions on Procedural and Cross-Cutting Issues, Note by the Secretariat' (8 July 2024) UN Doc A/CN.9/WG.III/WP.244 (Draft Provisions on Procedural and Cross-Cutting Issues) 11 (draft provision 19). For widely cited arbitral formulations, see, eg, *SD Myers, Inc v Canada*, UNCITRAL, Partial Award (13 November 2000) para 263. For IIAs that attempt to codify a notion of deference, see, eg, India–Belarus BIT, art 23.1.

¹⁰⁹ See, eg, Draft Statute of a Standing Mechanism (n 1) arts 27, 29; Kucik and Puig (n 47) 575–6; Kaufmann-Kohler and Potestà (n 17) para 118.

See Draft Statute of a Standing Mechanism ibid, arts 26(1), 36(1); Bungenberg and Reinisch (n 2) 75 (art 56(1)).
 See, eg, Bungenberg and Reinisch ibid 34.
 See, eg, Bungenberg and Reinisch (n 16) 161; Draft Statute of a Standing Mechanism (n 1)

arts 26(3), 36(3); New York Convention (n 55) art I.

¹¹³ See, eg, Bungenberg and Reinisch (n 2) 76 (arts 56(4) and 57); UNCITRAL, 'Possible Reform of Investor–State Dispute Settlement (ISDS): Appellate Mechanism and Enforcement Issues, Note by the Secretariat' (12 November 2020) UN Doc A/CN.9/WG.III/WP.202, para 44.

decisions—and undermine the effectiveness of a MIC or MAM. Accordingly, this would not be an advisable design choice. In contrast, if the adoption procedure were modelled on the 'reverse consensus' requirement in WTO dispute settlement—requiring consensus to block report adoption—it is unlikely that such a requirement would operate as a significant means of State control. Another option would be to enable some specified proportion of the Contracting Parties to block the adoption of decisions, or to determine that an interpretation shall not operate as a future precedent—ideas that have previously been proposed in relation to WTO dispute settlement.¹¹⁴

Another set of design questions concerns how the States that create a MIC or MAM may regulate the remedies that the tribunal may award. For example, following existing IIA practice, ¹¹⁵ it is likely that States may limit a MIC or MAM to awarding monetary damages or restitution of property with an alternative of paying damages in lieu of restitution, and prohibit the tribunal from awarding punitive damages. ¹¹⁶ More interestingly, a MIC or MAM statute might also place restrictions on how the tribunal may calculate any damages. For example, given existing concerns about this aspect of investor–State arbitration, a MIC or MAM statute might provide that the tribunal 'shall only award monetary damages that are established on the basis of satisfactory evidence and that are not inherently speculative' and set out the factors that the MIC or MAM must consider in assessing any damages. ¹¹⁷ Overall, it would seem advisable to include such limitations in a MIC or MAM statute.

IV. BILATERAL OR UNILATERAL CONTROL MECHANISMS

This section considers those control mechanisms that will remain exercisable jointly by the parties to the particular IIA at issue or that will be exercised unilaterally by individual parties to an IIA. The analysis in this section supports the assertion in this article that despite the procedural law in a MIC

¹¹⁴ See, eg, CE Barfield, Free Trade, Sovereignty, Democracy: The Future of the World Trade Organization (AEI Press 2001) 127 (proposing enabling one-third of WTO members, representing at least one-quarter of trade amongst WTO members, to block the adoption of panel or AB reports, or alternatively to allow the decision to stand for the particular dispute but not as a precedent); W Zhou and H Gao, ""Overreaching" or "Overreacting"? Reflections on the Judicial Function and Approaches of WTO Appellate Body' (2019) 53 JWT 951, 974–5 (suggesting enabling a respondent to seek a declaration that an issue is non-justiciable, or that an interpretation only applies to the case at hand, if supported by one-third of WTO members).

¹¹⁵ See above n 52.

¹¹⁶ See Draft Provisions on Procedural and Cross-Cutting Issues (n 108) 11–12 (draft provision 20(1) and (4)); Bungenberg and Reinisch (n 16) 150, 153–4.

¹¹⁷ See, eg, Draft Provisions on Procedural and Cross-Cutting Issues ibid 11–12 (draft provision 20(3)–(4)). See discussion in Report of Working Group III Forty-Ninth Session (n 1) paras 99–104, esp para 101; 2021 Canada Model Foreign Investment Promotion and Protection Agreement (FIPA) art 40(5)–(7).

or MAM having been multilateralized, certain key control mechanisms will remain exclusively in the hands of the parties to the relevant IIA.

A. Authoritative Interpretations, Treaty Amendments and Treaty Termination

The power to adopt authoritative interpretations of an IIA, including in response to prior decisions of a MIC or a MAM, will almost certainly remain solely in the hands of the parties to the particular IIA. This reflects the fact that provisions that are common in newer IIAs, which permit the treaty parties to adopt an interpretation of the agreement that binds investor—State arbitral tribunals or other dispute settlement bodies constituted under the treaty, require the agreement of all treaty parties. Similarly, subsequent agreement and subsequent practice of the parties to a treaty, pursuant to Articles 31(3)(a) and (b) of the Vienna Convention on the Law of Treaties (VCLT), require 'a common understanding regarding the interpretation of a treaty which the [treaty] parties are aware of and accept'. Likewise, the power to amend an IIA, including in response to prior decisions of a MIC or MAM, will remain exclusively in the hands of the parties to the particular IIA concerned. The power to terminate an IIA jointly, and neutralize the effects of any survival clause, will also remain solely in the hands of the parties to the relevant IIA.

While the power to adopt authoritative interpretations of an IIA will almost certainly remain a decision in the hands of the parties to the relevant IIA, as explained further below in Section V.D, in designing a MIC or MAM, States should create a formal process whereby all Member States of the MIC or

(n 16) 41.

119 See, eg, EU-Singapore IPA (n 2) arts 3.13(3), 4.1(4)(f), 4.2(3); CETA (n 2) arts 8.31(3), 8.44(3)(a), 26.1(5)(e), 26.3(3); Association of Southeast Asian Nations (ASEAN) Comprehensive Investment Agreement, art 40(2)–(3); DR-CAFTA (n 52) arts 10.22(3), 19.1(3)(c), (5).

¹²⁰ International Law Commission, Draft Conclusions on Subsequent Agreements and Subsequent Practice in relation to the Interpretation of Treaties with Commentaries [2018] II(2) UNYBILC 25, 63, Conclusion 10(1). See also commentary to Conclusion 10, paras 1–2 and commentary to Conclusion 4, esp paras 4, 9–12; Vienna Convention on the Law of Treaties (adopted 23 May 1969, entered into force 27 January 1980) 1155 UNTS 331 (VCLT) art 31(3)(a)–(b).

Provisions on amendments in IIAs typically require the agreement of the treaty parties and many require that the treaty parties complete their domestic procedures for ratification. See, eg, EU–Singapore IPA (n 2) art 4.3(1); CETA (n 2) art 30.2(1).

While IIAs often provide for the option of unilateral termination by one of the treaty parties, unilateral termination has the major downside that it is not possible to modify the survival clause in the treaty. In contrast, in the case of IIAs terminated by agreement of the treaty parties there are examples of agreements to alter the effects of a survival clause: see, eg, T Voon and AD Mitchell, 'Denunciation, Termination and Survival: The Interplay of Treaty Law and International Investment Law' (2016) 31 ICSID Rev 413, 423–31. Art 54 of the VCLT (n 120) provides that a treaty may be terminated either: '(a) in conformity with the provisions of the treaty; or (b) at any time by consent of all the parties after consultation with the other contracting States' (emphasis added).

¹¹⁸ See, eg, Submission from the European Union (n 2) paras 26–27; Bungenberg and Reinisch (n 16) 41.

MAM could propose, or register support for, written counter-interpretations issued in response to decisions of the tribunal, which could be applied by agreement to other IIAs besides the treaty invoked in the original proceedings. As Mehranvar and Johnson have suggested, there are also other reforms that States could consider adopting if they want to strengthen authoritative interpretations and subsequent agreement as tools of interpretative control. These could include providing that a non-disputing treaty party's silence shall be interpreted as agreement with the host State's position or strengthening the obligation of non-disputing treaty parties 'to engage constructively with their treaty partner(s)' over issues of treaty interpretation. 124

B. Control over the Jurisdiction of a MIC or MAM

Another control mechanism that will operate at a more localized level is the power to decide which categories of disputes, under which IIAs (or other instruments), the tribunal is given jurisdiction over. Obviously, all States involved in negotiating the statute of a MIC or MAM will be involved in negotiating the provisions regulating the tribunal's jurisdiction. However, such provisions are likely to leave most of the detail—in terms of opting-in to jurisdiction—to be determined by individual States Parties to the MIC or MAM. For example, in the UNCITRAL secretariat Draft Statute of a Standing Mechanism, the provisions on the jurisdiction of the first-instance and appellate tribunals leave the power to grant these bodies jurisdiction up to individual Contracting Parties. Specifically, the provisions provide for jurisdiction where the disputing parties have agreed in writing to submit the dispute to the relevant tribunal and also permit a Contracting Party to the Standing Mechanism to consent to the jurisdiction of either tribunal by providing a list of treaties or foreign investment laws over which it accepts the tribunal's jurisdiction. 125 Some States have suggested that the MIC or MAM's jurisdiction should be limited to instances where all parties to an IIA have accepted the tribunal's jurisdiction over claims under the IIA, to ensure consistency of treaty interpretation. 126 Even if this were made a requirement, it would mean that the decision whether to grant the MIC or MAM

¹²³ See Section V.D, text at nn 169-174.

¹²⁴ See L Mehranvar and L Johnson, 'Missing Masters: Causes, Consequences and Corrections for States' Disengagement from the Investment Treaty System' (2022) 13 JIDS 264, 288.

¹²⁵ Draft Statute of a Standing Mechanism (n 1) arts 14(1)–(2), 18(1)–(2). Note that a Contracting Party can limit its consent to jurisdiction to disputes arising under the list of instruments provided and can also limit its consent to instances where the claimant is a national of a Contracting Party: see art 39(1)(a), (c).

¹²⁶ See Submission of Canada on the Informal Documents prepared by the Secretariat to Facilitate Discussions at the Sixth Intersessional Meeting of Working Group III (January 2024) 3 https://uncitral.un.org/sites/uncitral.un.org/files/media-documents/uncitral/canadas_comments_january_2024.pdf. See also Report of Working Group III Forty-Eighth Session (n 1) para 99.

jurisdiction would rest with two States (typically the parties to a bilateral IIA)—ie still only a small subset of the parties to a MIC or MAM.

V. BALANCING THE INTERESTS OF THE PARTIES TO AN IIA WITH SYSTEMIC INTERESTS

This section considers areas where there is a tension between the interests of the parties to an IIA in controlling the interpretation of their treaty, and the wider systemic interests of other Member States of the MIC or MAM that may be affected by proceedings before the tribunal. It argues that third-party participation in proceedings before a MIC or MAM—both by States that are parties to the IIA invoked but non-parties to the MIC or MAM, and by States that are parties to the MIC or MAM but not parties to the relevant IIA—is an area where the tension between procedural multilateralism and substantive bilateralism will be particularly acute. Besides third-State participation, this section also considers any right that is created to comment on draft awards of a MIC or MAM, or any forum that is created for MIC or MAM Member States to discuss, and potentially override, the tribunal's interpretations, which have been suggested as potential control mechanisms and would also raise the tension between substantive bilateralism and procedural multilateralism.

A. Participation by Non-Disputing Parties to the IIA

A strong case can be made for the right of States that are parties to the IIA at issue in proceedings before a MIC or MAM to make submissions on issues of treaty interpretation. Numerous IIAs already provide for non-disputing treaty parties to make submissions regarding issues of treaty interpretation in investor–State proceedings under the IIA, 127 and Article 5(1) of the UNCITRAL Rules on Transparency in Treaty-based Investor–State Arbitration also recognizes such a right. 128 While these existing bases for non-disputing Party participation are drafted with investor–State arbitration in mind, there is no principled reason why they should not also apply to proceedings before a MIC or MAM. Furthermore, it is likely that the MIC or MAM's statute will provide that the tribunal shall follow the UNCITRAL Transparency Rules. 129

One issue that will arise—assuming it is not made a jurisdictional requirement that all parties to an IIA are parties to the MIC or MAM—is that the non-disputing treaty party may be a party to the relevant IIA relied upon for State consent in proceedings before a MIC or MAM but not be a party to the MIC or MAM's statute. A State's non-participation in the MIC or MAM

¹²⁸ UNCITRAL Rules on Transparency in Treaty-based Investor–State Arbitration (UNCITRAL Transparency Rules) art 5(1).

See Draft Statute of a Standing Mechanism (n 1) art 22(4).

 $^{^{127}}$ See, eg, NAFTA (n 52) art 1128; DR-CAFTA (n 52) art 10.20(2). On the use of these provisions, see Polanco (n 12) 173–86; Mehranvar and Johnson (n 124) 276–80.

does not seem to be a good reason for denying that State the right to make a non-disputing Party submission when an IIA to which it is party is at issue before either body. While there may be a concern that non-parties to the MIC or MAM would not have contributed to the tribunal's budget, but would have some involvement in the proceedings (in making non-disputing Party submissions), 130 such concerns could be managed by orders setting the parameters for such participation, eg requiring the non-member of the MIC or MAM to pay a proportion of costs attributable to its participation. 131

B. Third-Party Participation by States that are Parties to the Statute but not the IIA

The more challenging question in terms of third-party participation is whether States that are parties to the statute of the MIC or MAM, but are not parties to the IIA that is invoked as a basis of the tribunal's jurisdiction, should have any right to participate in proceedings. There are at least two reasons why such a third State may need to be afforded some right to participate in the proceedings. First, it is likely that litigation will arise over the interpretation of the statute of a MIC or MAM and the tribunal's procedural rules. Arguably all States Parties to a MIC or MAM would have a legitimate interest in how such questions are resolved and hence may need to be afforded a right to make submissions, despite not being a party to the IIA invoked as a basis of jurisdiction. Second, as investment treaties contain many broadly similar concepts and standards, ¹³² and as one of the motivations for creating a MIC or MAM is to bring greater consistency of approach to the interpretation of IIAs, 133 it is foreseeable that positions the tribunal takes in proceedings in relation to one IIA may have significant implications for the IIAs of other State Parties to the MIC or MAM, as such a body is likely to find its own case law persuasive in later disputes. 134 Indeed, Bungenberg and Reinisch suggest that a MIC might be instructed in its statute to prefer interpretations that favour consistency and coherence in the interpretation of IIAs, a step that would obviously increase the systemic importance of its decisions. 135

¹³⁰ See generally Bungenberg and Reinisch (n 16) 65, 94.

See, eg, UNCITRAL Transparency Rules (n 128) art 5(4); ibid 94.

¹³² See, eg, SW Schill, 'Ordering Paradigms in International Investment Law: Bilateralism—Multilateralism—Multilateralization' in Z Douglas, J Pauwelyn and JE Viñuales (eds), *The Foundations of International Investment Law: Bringing Theory into Practice* (OUP 2014) 114–17, 138.

Bungenberg and Reinisch (n 16) 126–7; Bungenberg and Reinisch (n 2) 62 (draft article 28(1)(d)). See also the far stronger suggestion of Florou, who argues that the States that create a MIC should delegate to it the power to develop subsequent practice pursuant to IIAs that the MIC is given the power to interpret, and that this might also apply to the IIAs of other MIC members if they do not react with a specified period: A Florou, 'Multilateralising Interpretation: Fitting the Rules of the VCLT into the Multilateral Investment Court (or Vice Versa)' in E Shirlow and KN Gore (eds), *The Vienna Convention on the Law of Treaties in Investor–State Disputes: History, Evolution and Future* (Kluwer Law International 2022) 344–7.

A good example of an issue of systemic importance for numerous IIAs that will very likely come before a MIC or MAM is the content of the customary international minimum standard of treatment (MST) and whether that standard has evolved from the seminal *Neer* formulation with which it is often equated. ¹³⁶ While the wording of an applicable IIA would influence precisely how the issue of the MST is relevant, the MIC or MAM's views on the content of the MST, which it could be expected to find persuasive in later cases, would have systemic relevance for all Parties to the MIC or MAM, including those not party to the IIA invoked in the initial case. ¹³⁷

The question then becomes, should States that are parties to the statute of the MIC or MAM, but not parties to the relevant IIA at issue in a case, have some right to make submissions where the proceedings raise issues of systemic importance, with relevance for many other IIAs? The EU's main submission to Working Group III suggested that:

it should be considered whether and, if so, under what conditions other governments that are party to the instrument establishing the standing mechanism should be able to intervene in disputes on questions of interpretation of systemic importance under treaties to which they are not contracting parties, while ensuring at the same time that this does not compromise the ability of the parties to an agreement to retain control over its interpretation. ¹³⁸

Bungenberg and Reinisch also suggest that a MIC's statute or procedural rules should provide that 'a MIC Member who demonstrates a legal interest in a pending dispute can be admitted by the MIC as an intervening third party'. ¹³⁹ The reference to a 'legal interest' appears inspired by Article 62 of the ICJ Statute which provides for intervention by a third State where the State has 'an interest of a legal nature which may be affected by the decision in the case'. ¹⁴⁰

The problem with a procedure for third-party participation before a MIC or MAM modelled on Article 62 of the ICJ Statute is that it would not extend to providing a sufficiently broad right of participation for a State that is a party to the MIC or MAM, but not a party to the IIA invoked in a case. The requirement in Article 62 that a State must have 'an interest of a legal nature which may be affected by the decision in the case' has been interpreted restrictively, and it is

¹³⁶ L.F.H. Neer and Pauline E. Neer v Mexico, US-Mexico Claims Commission, Opinion (15 October 1926) IV RIAA 60, para 4. For a sense of this issue see, eg, P Dumberry, 'Fair and Equitable Treatment: Its Interaction with the Minimum Standard and Its Customary Status' (2017) 1 BrillResPerspIntlInvL&Arb 1, 3–46.

¹³⁷ Consider: M Jarrett, 'ISDS 2.0: Time for a Doctrine of Precedent?' (2024) 27 JIEL 41, 43, 47 ("adjudicative law in ISDS 2.0" will [include] ... pronouncements on customary international law and general principles of law'); AK Bjorklund, 'Arbitration, the World Trade Organization, and the Creation of a Multilateral Investment Court' (2021) 37 ArbIntl 433, 442, noting some investment protection obligations are found in customary international law but emphasizing that 'some states have deliberately taken a different path in certain areas'.

Submission from the European Union (n 2) para 27.

Bungenberg and Reinisch (n 16) 97.

¹⁴⁰ ICJ Statute (n 43) art 62(1).

well established in the ICJ's jurisprudence that a general interest of a third State in the law to be applied by the Court is not sufficient. ¹⁴¹

A more promising model to consider in designing a MIC or MAM may be the approach developed in WTO dispute settlement to third-party participation by WTO members besides the disputing parties. The DSU provides that any WTO member 'having a substantial interest in a matter before a panel and having notified its interest to the DSB ... shall have an opportunity to be heard by the panel and to make written submissions to the panel'. 142 The requirement of a 'substantial interest' in a matter before a panel has not been interpreted in a manner that restricts the ability of other WTO members to intervene and. as Iwasawa notes, '[a] general interest in the interpretation of the WTO Agreement or a systemic interest in the dispute settlement procedures is considered sufficient'. 143 A good example is the panel proceedings in the *United* States - Section 301 Trade Act case, where the EU's claims largely concerned the consistency of the United States' (US) trade laws with the WTO DSU—ie litigation over the constitutive instrument—and some 11 WTO members made submissions as third parties that largely focused on the systemic implications of the dispute for the WTO dispute settlement system.¹⁴⁴ In drawing on WTO practice regarding third-party participation it must be remembered that unlike the WTO, even under a MIC or MAM the investment treaty regime will not feature uniform treaty text on issues beyond dispute settlement. 145 This may mean it is appropriate to develop certain additional limitations on third-party participation by a State that is party to the MIC or MAM statute but not the IIA invoked—eg perhaps requiring such a State to show that it has at least one IIA with identical wording to the IIA at issue in the proceedings in which it wishes to make a submission. 146 Were such a limitation introduced, further thought would be needed regarding how it might apply when the focus of the proposed intervention was a point of customary international law, or a general principle of

¹⁴¹ See, eg, Y Iwasawa, 'WTO Dispute Settlement as Judicial Supervision' (2002) 5 JIEL 287, 301; Land, Island and Maritime Frontier Dispute (El Salvador/Honduras) (Application to Intervene, Judgment) [1990] ICJ Rep 92, para 76; Continental Shelf (Tunisia/Libyan Arab Jamahiriya) (Application to Intervene, Judgment) [1981] ICJ Rep 3, para 30; R Kolb, The International Court of Justice (A Perry trans, Hart Publishing 2013) 709–12.

¹⁴² WTO DSU (n 26) art 10(2).
143 Iwasawa (n 141) 302–3; Carmody demonstrates that the notion of a 'substantial interest' was also interpreted in permissive terms in GATT-era dispute settlement: C Carmody, 'Of Substantial Interest: Third Parties under GATT' (1997) 18 MichJIntIL 615, 624–35. Note that 'enhanced third-party rights', which go beyond those provided for in the DSU, are often requested in WTO panel proceedings and there is an established practice of granting them in some circumstances: see generally T Sekine, 'Enhanced Third Party Rights under the WTO Dispute Settlement System' (2018) 15 ManchesterJIntIEconL 354 Sekine notes that enhanced third-party rights have not been granted merely on the basis of systemic interests in the dispute: 372–3.

¹⁴⁴ See generally Panel Report, *United States – Sections 301–310 of the Trade Act of 1974*, WT/DS152/R, adopted 27 January 2000, paras 5.1–5.361, 7.1–7.8.

Similarly, Kucik and Puig (n 47) 571–2.
 Summary of the Inter-Sessional Meeting Submitted by the Government of Singapore (n 4) para 62.

law, which is in turn relevant to the interpretation of IIAs, eg the question of the content and evolution of the MST mentioned above.¹⁴⁷

Ultimately, there is no easy answer to what 'level of interest' should be required for a third State, which is a party to the MIC's or MAM's statute but not the relevant IIA invoked, to intervene in proceedings. 148 Any right afforded to such States would need limits placed upon it, so as to avoid unduly expanding and burdening the proceedings before the tribunal. 149 Before WTO panels, third parties that notify their interest receive the submissions of the parties to the first panel meeting, and are given an opportunity to make written submissions to the panel and to be heard by the panel at a session of the first panel meeting dedicated to hearing such third-party views. 150 The GATT/WTO practice appears to be to disregard issues that are only raised by third parties and not the disputing parties. 151 This may also be an appropriate limit in proceedings before a MIC or MAM, to prevent third parties which are only parties to the statute but not the specific IIA invoked from unduly expanding the proceedings. Much like existing rules for submissions by non-disputing treaty parties in investor-State arbitration, a MIC or MAM would have the power to regulate the extent of submissions of other States Parties to the MIC or MAM to ensure fairness to the disputing parties. 152

Some readers might question whether States that are parties to the MIC or MAM statute but not to the IIA invoked as a basis for jurisdiction should have any right to intervene on issues of systemic importance (eg concerning the construction of the MIC or MAM statute, or concerning standards or concepts common to numerous other IIAs). As noted above, in the ICJ and ITLOS, a general interest in the interpretation of the Statute of the Court or Tribunal and its procedural rules, or a general interest in the legal principles to be applied, is clearly not sufficient to warrant intervention, ¹⁵³ even though the Court or Tribunal is very likely to follow its interpretation in later cases. In the author's view, these examples are a less appropriate model for third-State participation before a MIC or MAM than the GATT/WTO approach to third-party participation for at least two reasons. First, a MIC or MAM is likely be a tribunal with a high volume of litigation, where States are largely respondents who bear most of the litigation risk, thereby increasing the

¹⁴⁷ See text accompanying nn 136–137.

¹⁴⁸ See the discussion of relevant policy considerations (focusing on the GATT context) in Carmody (n 143) 621.

¹⁵⁰ WTO DSU (n 26) art 10(2)–(3) and Appendix 3, para 6; D Palmeter, PC Mavroidis and N Meagher, *Dispute Settlement in the World Trade Organization: Practice and Procedure* (3rd edn, CUP 2022) 223–4.

edn, CUP 2022) 223-4.

151 Carmody (n 143) 632-3. Panel Report, European Communities – Protection of Trademarks and Geographical Indications for Agricultural Products and Foodstuffs, Complaint by the United States, WT/DS174/R, adopted 20 April 2005, para 7.35 (refusing to make a recommendation requested by Mexico, as a third party, that related to an issue that was not challenged by the claims in dispute and was therefore outside the panel's terms of reference).

See, eg, UNCITRAL Transparency Rules (n 128) art 5(4).

See text accompanying n 141.

systemic interest of other Member States in how the MIC or MAM's statute is interpreted. Second, the substantial similarities in the standards and concepts found in numerous IIAs means that it is artificial to pretend that other members of a MIC or MAM will not have a significant interest in how the MIC or MAM interprets such standards and concepts, given the likely impact on later disputes under other IIAs in which the MIC or MAM can be expected to find its earlier interpretation influential. 154

Finally, and most challengingly, it can be asked whether States that are not parties to the MIC or MAM statute or the IIA invoked as a basis for the tribunal's jurisdiction should be afforded any right to make a third-party submission given the potential for the MIC or MAM's interpretation of relevant terms or concepts to influence later decisions of investor-State arbitral tribunals. In this scenario, one view would be there is no convincing case for providing a right to make a third-party submission because there is nothing to link such non-parties to the MIC or MAM (unlike the situation addressed above where an IIA to which they are a party comes before the MIC or MAM). A contrasting view would stress that the MIC or MAM's decisions are likely to prove influential before subsequent investor-State arbitral tribunals, potentially having some sort of superior quality, thus weighing in favour of some ability for such non-parties to make a submission. 155 Ultimately, it seems unlikely that the States that create a MIC or MAM would be willing to provide a basis for non-parties to the tribunal to participate in cases beyond the limited scenario where the non-party is a party to the IIA relied on as a basis for jurisdiction, given that such involvement would extend proceedings and undermine incentives to join the MIC or MAM. 156

C. A Right to Comment on Draft Awards

A further control mechanism where there could be a tension between the interests of the treaty parties or the disputing parties and the wider systemic interests of all Member States of a MIC or MAM concerns any right created to comment on draft awards. Such a right is provided for in a small number of newer IIAs, which, following US IIA practice, establish that, at the request of either disputing party, an investor–State arbitral tribunal shall circulate its proposed decision on liability to the disputing parties and the non-disputing treaty party or parties. Under these provisions, the disputing parties—but not the non-disputing treaty party or parties—are given an opportunity to

¹⁵⁴ On the potential for a MIC or MAM to develop coherent interpretations of common terms in IIAs, while giving weight to differences in treaty wording, see Kaufmann-Kohler and Potestà (n 17) paras 73, 188; AK Bjorklund, 'The Road(s) Not Taken—the Past, Present, and Future of International Investment Law Reform' (2023) 39 ArbIntl 455, 466.

¹⁵⁵ See, eg, 6th Intersessional Meeting of UNCITRAL Working Group III, September 2023, panel 4 at 32.10–34.40 (remarks of Karin Kizer) https://youtu.be/wDfcvM8x4v4.

¹⁵⁶ See also Bungenberg and Reinisch (n 16) 64–5 (on creating incentives to join a MIC).
157 eg US–Singapore Free Trade Agreement, art 15.19(9)(a); DR-CAFTA (n 52) art 10.20(9)(a).
In some treaties the obligation is only to circulate the draft award to the disputing parties: see, eg,

submit written comments to the tribunal concerning any aspect of its proposed award, which the tribunal is obliged to consider before issuing its final award. 158 One of the purposes of such procedures is 'to permit both treaty Parties an opportunity to make their views known as to the impact of a proposed award on issues of public interest'159 and issues of treaty interpretation, and, potentially, to coordinate and express a joint view disapproving of the tribunal's interpretation. 160 Coe highlights that the treaty parties may react to the circulation of a draft award containing an interpretation of an IIA they disagree with by adopting an authoritative interpretation that binds the tribunal. 161 Several commentators have highlighted a right to comment on draft awards as a control mechanism that might be further developed in the investment treaty regime, including in a standing appellate mechanism. 162 For example, Yackee suggested a stronger version of such a mechanism may require automatic circulation of draft awards to all treaty parties and permit non-disputing treaty parties to submit comments, ¹⁶³ or permit the parties to an IIA to veto jointly law-making aspects of an award with which they disagree. 164

Were such a right to comment on draft awards created, arguably there could be a rationale for extending it to all parties to a MIC or MAM for the reasons developed above—ie, that a MIC or MAM's decisions may concern the interpretation of the tribunal's statute and procedural rules or address issues with systemic relevance for numerous other IIAs. 165 However, such a right to comment on draft awards sits uneasily with judicial independence and accordingly it does not seem advisable to include this control mechanism in the statute of a MIC or MAM. This is because the aim of submitting comments on draft awards would be to alter the reasoning or conclusions in draft awards after the tribunal has engaged in its deliberations. In contrast, third-party submissions made before the tribunal has engaged in its deliberations, considered above, do not raise the same concerns regarding judicial independence. 166

Comprehensive and Progressive Agreement for Trans-Pacific Partnership, art 9.23(10); Indonesia-Singapore BIT, art 24(4).

MA Kantor, 'The New Draft Model U.S. BIT: Noteworthy Developments' (2004) 21 JIntlArb

383, 390.

160 LM Caplan and JK Sharpe, 'United States' in C Brown (ed), *Commentaries on Selected* Model Investment Treaties (OUP 2013) 836; JJ Coe Jr, 'An Examination of the Draft Award Circulation Provision of the US Model BIT of 2004' in CA Rogers and RP Alford (eds), The 161 Coe ibid 117–18, 123. Future of Investment Arbitration (OUP 2009) 117-18.

162 Kucik and Puig (n 47) 577.

JW Yackee, 'Controlling the International Investment Law Agency' (2012) 53 HarvIntlLJ See ibid 443–4. 391, 438-9.

Although note that in WTO panel proceedings, in the interim review stage the right to comment is only afforded to the parties to the dispute, not to third parties: WTO DSU (n 26) art 15. A Mastromatteo and S Sinha, 'Interim Review: Dispute Settlement System of the World Trade Organization (WTO)' in H Ruiz Fabri (ed), Max Planck Encyclopedia of International Procedural Law (OUP 2018) paras 12, 15.

¹⁶⁶ See, eg, Columbia Center on Sustainable Investment, video of seminar on 'Interpretation of Treaties', 16 November 2022, at 53:30–55:10 (comments of Kabir Duggal and Arianna Arce) ">.

D. A Forum for Member States to Discuss Awards and Propose Counter-Interpretations

Rather than a right to comment on draft awards, an alternative mechanism to give States Parties to a MIC or MAM an opportunity to provide periodic input on the tribunal's jurisprudence may be to create a dedicated forum for this purpose, which is disconnected from the adjudicatory process in specific disputes. For example, in the WTO context, various proposals have been made to enable members to voice their views on adjudicatory decisions in a manner that is detached from report adoption in specific disputes, eg via an annual meeting of the DSB for this purpose. 167 Compared to a right to comment on a draft award, the advantage of a periodic forum for members to comment on the MIC or MAM's jurisprudence and operation is that it would be less clearly aimed at asking adjudicators to reverse their decision or reasoning in a pending case. However, there would still be significant tension between the influence of such a political forum and the value of judicial independence. It may also be unclear what (if anything) a MIC or MAM should do in response to comments of States Parties, for example where there are a variety of views expressed. 168

A related, more advisable idea would be to create a formal procedure whereby the States Parties to a MIC or MAM could propose, and indicate support for, written counter-interpretations that could be issued in response to the new standing tribunal's interpretations. In the WTO context, Fukunaga has suggested creating a formalized procedure whereby members could circulate written interpretative declarations in response to adjudicatory interpretations, essentially as a 'counter-proposal'. 169 Some such statements might achieve sufficient acceptance to constitute subsequent agreement among the parties to an IIA, or alternatively have more limited weight as a supplementary means of interpretation. ¹⁷⁰ Jarrett has in rough terms suggested something similar in the context of a MIC or MAM, arguing there 'should be a multilateral forum [created] for treaty parties to annually discuss adjudicative law produced by the appellate tribunal' and to propose changes to the tribunal's precedents, which would apply to the IIAs between all States that support the proposed change to the adjudicative precedent. 171 What these two proposals highlight is that it is important to go beyond merely providing for a regular forum in

¹⁷¹ Jarrett (n 137) 50.

¹⁶⁷ Paine (n 56) 851–2.

See, eg, J Pauwelyn, 'The Rule of Law without the Rule of Lawyers? Why Investment Arbitrators Are from Mars, Trade Adjudicators from Venus' (2015) 109 AJIL 761, 802–3.
 Y Fukunaga, 'The Appellate Body's Power to Interpret the WTO Agreements and WTO

Members' Power to Disagree with the Appellate Body' (2019) 20 JWorldInv&Trade 792, 812–17. 170 ibid 815–17. Agreements between the parties to an IIA reached within such a written procedure tied to a MIC or MAM would not necessarily satisfy the requirements of IIA provisions that empower the treaty parties to adopt binding joint interpretations, as the latter provisions often have specific procedural requirements: see, eg, the treaties cited in n 119.

which MIC or MAM Member States can discuss the tribunal's jurisprudence, through the creation of a procedure whereby Member States can propose, and indicate support for, counter-interpretations that effectively respond to the tribunal's interpretations. Such a procedure would amount to creating a centralized, structured process for treaty parties to do something they are already entitled to do, namely publish post-adjudication interpretative statements, including in a coordinated manner. As the plenary forum in which such counter-proposals would be considered would involve all States Parties to the MIC or MAM, it would recognize the reality, emphasized above, that the issues raised may be of systemic importance for the IIAs of numerous States, while still leaving each Member State to decide whether to apply a proposed counter-interpretation to its IIAs. 172 If such a mechanism were created, it would be important to consider how written counterinterpretations, once adopted by two or more Member States, would apply to pending disputes, as permitting counter-interpretations to apply to disputes at an advanced stage would raise concerns about judicial independence and States being permitted to adopt retroactive amendments. 173

Ultimately, this proposed mechanism for circulation of, and registering support for, written counter-interpretations in response to decisions of a MIC or MAM, would create a meaningful avenue for State input and control without introducing significant risks of unilateral vetoes or abuse. This is because a written counter-interpretation would only constitute a subsequent agreement where two or more Contracting Parties agreed to apply it to the IIA(s) between them, whereas a counter-interpretation only supported by an individual treaty party would receive limited (if any) interpretative weight. Also, as noted above, limits would be placed on applying counter-interpretations retrospectively to disputes at an advanced stage. This proposal for creating a process for States to issue written counter-interpretations is consistent with States' broader efforts within the Working Group III process to create 'tools through which they can monitor and manage the investment treaty system to shape its evolution and emergence', which will outlive the UNCITRAL process.¹⁷⁴

VI. CONCLUSION

This article has analysed the design of control mechanisms in a future MIC or MAM. A MIC or MAM will operate on a plurilateral basis, among States that

¹⁷² See generally ibid.

¹⁷³ See, eg, Roberts and St John (n 11) 139; A Roberts, 'Power and Persuasion in Investment Treaty Interpretation: The Dual Role of States' (2010) 104 AJIL 179, 210–14 (discussing the issue of timing in relation to treaty parties' joint interpretations). 2019 Netherlands Model BIT, art 24(2) (providing that joint interpretations adopted by the treaty parties are not applicable in cases where an investor has already submitted a claim).

¹⁷⁴ Roberts and St John (n 11) 132, 127–8, 142–3.

choose to become parties to the tribunal's constitutive instrument and grant it jurisdiction over disputes under their IIAs. This article has highlighted two fundamental tensions that underlie and animate the many specific questions of institutional design. The first is the tension between independence and accountability, which is relatively familiar from other international tribunals. The second is the tension between procedural multilateralism and substantive bilateralism, which arises because a MIC or MAM will only involve multilateral agreement over dispute settlement procedures, and the relevant substantive law will remain contained in mostly bilateral investment treaties, controlled by the parties to those agreements. As this article has demonstrated, while certain key control mechanisms will continue to operate at a bilateral level, in the hands of the parties to the relevant IIA—eg authoritative interpretations, treaty amendments—others will be exercised by all parties to a MIC or MAM. The article has argued that while a MIC or MAM should include adequate mechanisms for Member State control, given the increased interpretative influence of the new centralized tribunal, it is important that such control mechanisms are designed so that they will not unduly undermine the independence of the tribunal.

A core contribution of this article has been to work though the tension between procedural multilateralism and substantive bilateralism which will arise in a more acute manner with the development of a MIC or MAM. In short, under a MIC or MAM, while the procedural law will have been multilateralized (in the tribunal's statute and procedural rules), the substantive law will remain largely bilateral. However, a MIC or MAM's decisions will be likely to have significant implications for numerous IIAs beyond the specific treaty at issue, given the similarity in content across many IIAs, the potential of the MIC or MAM to find its own interpretative approach persuasive in later cases, 175 and the increased weight the new centralized tribunal's decisions can be expected to carry. 176 This article has provided answers to consequent questions of institutional design, including how to regulate third-State participation in proceedings before a MIC or MAM—given the likely relevance of the tribunal's interpretations for numerous IIAs beyond the specific treaty at issue—and how to create a process whereby all MIC or MAM Member States could propose, or indicate support for, written counter-interpretations that respond to the tribunal's interpretations.

Besides addressing numerous questions of institutional design, this article has also provided a conceptual framework for understanding these issues that can assist policymakers considering the reform options of a MIC or MAM even if the specific approaches suggested here are not adopted. Through its analysis of control mechanisms in a MIC or MAM, this article also contributes to wider debates in international (economic) law over striking an appropriate balance between international judicial independence and Member

¹⁷⁵ See, eg, references cited in n 154.

State control. The article's analysis provides additional support for recent observations made by other commentators, who, reflecting on current ISDS reform processes, have variously noted that there are no perfect solutions in designing a MIC or MAM and all options involve trade-offs,¹⁷⁷ that the different reform options are interdependent,¹⁷⁸ and that 'it is ... proving difficult to move from a culture of bilateralism in investment arbitration to one of multilateralism'.¹⁷⁹

ACKNOWLEDGEMENTS

For helpful comments on prior drafts the author thanks an anonymous reviewer and numerous colleagues who offered comments within the 2023 ISDS Academic Forum conference, a 2023 workshop of the European Society of International Law (ESIL) Interest Group on International Courts and Tribunals, a 2023 session of the International Economic Law and Policy work in progress series, a 2023 seminar at the University of Bristol Law School, and a 2024 seminar in the London School of Economics public international law (LSE PIL) Hub seminar series.

¹⁷⁷ Kucik and Puig (n 47) 563, 568–9. 179 Bjorklund (n 154) 463.

¹⁷⁸ Roberts and St John (n 11) 114–15.